Time Recording Log, CS699 (Software Laboratory), Dept. of CS&E, IIT Bombay

Name : Roll No:

Login name (for email) :

Following are some examples of activities for which the time should be recorded. Feel free to include any other activity which requires time to be spent toward this course. Specify the program unit wherever applicable (in terms of module, program file, data structure/class, function etc.)

Reading/understanding the assignment statement Reading documents/books/manuals Discussing with others Designing overall solution (on paper) Designing the program on paper Actual Coding Time Testing/Debugging Modifying/Cleaning up the program

C? Stands for whether the work was completed or not.

Date Start Stop Interruption Time Activity Comments C?

Time Time Spent